Author's response to reviews

Title: Russian nursing students' knowledge level and attitudes in the context of HIV

Authors:

Tarja Suominen (tarja.suominen@uta.fi)
Laura Laakkonen (puolitaivallaura@hotmail.com)
Dmitry Lioznov (dlioznow@yandex.ru)
May Polukova (poluk74@mail.ru)
Svetlana Nikolaenko (nikolaenkoS@yandex.ru)
Liudmila Lipiäinen (liudmila.lipiainen@uta.fi)
Maritta Välimäki (maritta.valimaki@utu.fi)
Jari Kylmä (jari.kylma@uta.fi)

Version: 2 Date: 20 September 2013

Author's response to reviews: see over
Dear Editor

Attached, please find again our article “Russian nursing students’ knowledge level and attitudes in the context of HIV – a descriptive study” submitted for publication in BMC Nursing. We hope this article is of interest. Before, only limited number of articles has been published in English of the data collected in nursing area in Russia.

The content of this article has not been published before, neither submitted for publication elsewhere. All authors have contributed significantly and all authors agree with the content of the manuscript.

The language of this text has been corrected by an official translation office with native English speaking people.

Changes made: the study design has been added in the title, the correspondent author has been marked in the title page. The data were collected between May and April 2010 after receiving permission for the study from all study institutions according to their research standards, based on these standards the institution is offering the approval, not ethics committee when it is question of adult students. The adding has been made in the abstract text.

On behalf of all author, the chair and responsible person of this project

In Helsinki, Finland 20th September, 2013

Tarja Suominen, PhD
Professor
University of Tampere
School of Health Sciences, Nursing Science
Finland